

Vitamin D deficiency and associated factors among children in a tertiary care hospital setting: A retrospective cohort study

۲

Tzu-Ching Lin^{a,b,c}, Hsiao-Yun Yeh^{b,d,e}, Hsin-Lin Tsai^{b,f}, Pin-Hsuan Chiang⁹, Zih-Kai Kao^h, Ling-Yu Yang^{b,d}, Yu-Chun Chen^{b,g,i}, Jei-Wen Chang^{a,b,j,*}

^aDepartment of Pediatrics, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^bFaculty of Medicine, School of Medicine, National Yang Ming Chiao Tung University, Taipei, Taiwan, ROC; ^cDepartment of Pediatrics, China Medical University Hsinchu Hospital, Hsinchu, Taiwan, ROC; ^dDepartment of Medical Education, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^eDepartment of Family Medicine, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^(D)Division of Pediatric Surgery, Department of Surgery, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^eBig Data Center, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^hDepartment of Information Management, Taipei Veterans General Hospital, Taipei, Taiwan, ROC; ^ITaipei Veterans General Hospital, Yuli branch, Hualien, Taiwan, ROC; ^IInstitute of Emergency and Critical Care Medicine, School of Medicine, National Yang Ming Chiao Tung University, Taipei, Taiwan, ROC

Abstract

۲

Background: Vitamin D plays important roles in skeletal and extra-skeletal tissues. However, little is known about the vitamin D status in children in a hospital setting. We aimed to assess the vitamin D status, risk factors for vitamin D deficiency (VDD), and changes in biochemical profiles among children at a tertiary medical center in Taiwan.

Methods: This retrospective cohort study analyzed data from the Big Data Center of Taipei Veterans General Hospital. Children under 18-year-old who underwent 25-hydroxyvitamin D measurements between 2018 and 2023 were enrolled. Logistic regression analysis was performed to determine the risk factors for VDD.

Results: Among 1222 children enrolled, with a mean age of 8.5 ± 5.4 years, 597 (48.9%) had VDD. VDD was significantly associated with female sex (odds ratio [OR] = 1.624, 95% CI = 1.270-2.076), age >12 years (OR = 2.479, 95% CI = 1.584-3.881), vitamin D measurement during the spring/winter seasons (OR = 1.716, 95% CI = 1.340-2.197), and hospitalized children (OR = 1.949, 95% CI = 1.439-2.640). Age >1 to 6 years was a protective factor against VDD (OR = 0.391, 95% CI = 0.244-0.628). In addition, the OR of VDD was higher in those with an elevated intact parathyroid hormone level (OR = 8.667, 95% CI = 1.338-56.157). **Conclusion:** Despite the high sun exposure in Taiwan, VDD is prevalent among children and adolescents. Physicians should be aware of VDD, especially in children who are female, aged >12 years, hospitalized, have increased intact parathyroid hormone levels, and during the spring/winter.

Keywords: Adolescent; Child; Vitamin D; Vitamin D deficiency



www.ejcma.org

۲

Lin et al.

Lay Summary: Vitamin D deficiency (VDD) is a common medical problem worldwide. The prevalence of VDD in children varies among different geographical areas, ethnic and age groups, and the associations of biochemical parameters with VDD in children have yet to be clarified. This study investigated the epidemiology, clinical risk factors and biochemical profiles associated with VDD among Taiwanese children in a tertiary care hospital setting. The results showed that VDD was prevalent in Taiwanese children, and that those who were female, aged >12 years, hospitalized, had increased intact parathyroid hormone levels, and in whom vitamin D was measured during the spring/winter seasons were at a greater risk of VDD.

1. INTRODUCTION

Vitamin D plays a pivotal role in skeletal health, and emerging evidence suggests that it has effects on various extraskeletal tissues. Vitamin D may protect individuals from cancer through antiproliferative and prodifferentiating effects. In addition, vitamin D may prevent diabetes mellitus by modulating insulin secretion and peripheral actions, and it may also mitigate the risk of obesity. Furthermore, vitamin D has been shown to have a potential role in preventing cardiovascular diseases and modulating both adaptive and innate immune systems.^{1,2} Reported risk factors for vitamin D deficiency (VDD) include female sex, premature birth, dark skin pigmentation, inadequate dietary vitamin D intake, lack of sun exposure due to clothing or use of sunscreen, winter season, higher geographical latitude, insufficient daytime outdoor activities, obesity, certain medications (eg, anticonvulsants, glucocorticoids, antifungal agents), malabsorption, and liver and kidney diseases.2-5

Although clinical manifestations of severe VDD such as rickets in children are uncommon, a high prevalence of subclinical asymptomatic VDD has been reported worldwide, even in countries with high sun exposure.6-8 The prevalence of VDD in children varies from 33% to 90% depending on the cut-off values used to define VDD, geographical latitude, ethnicity, and age.6-10 Geographically, Taiwan is located in East Asia between 21°N and 25°N latitude. In the early 20th century, Taiwanese men were found to be 3 cm taller than men living in South Korea, which is also in East Asia. Although the per capita consumption of animal protein in Taiwan was 60% greater than that in South Korea in the 2000s, Taiwanese are now an average of 1 to 2 cm shorter than their South Korean counterparts. Other nutritional insufficiencies such as low levels of vitamin D can potentially cause impaired height growth in children. However, few large-scale studies have investigated the vitamin

*Address correspondence. Dr. Jei-Wen Chang, Department of Pediatrics, Taipei Veterans General Hospital, 201, Section 2, Shi-Pai Road, Taipei 112, Taiwan, ROC. E-mail address: jwchang@vghtpe.gov.tw (J.-W. Chang).

Journal of Chinese Medical Association. (2025) 88: 433-441.

Received September 13, 2024; accepted February 27, 2025.

Copyright © 2025, the Chinese Medical Association. This is an open access article under the CC BY-NC-ND license (http://creativecommons.org/licenses/ by-nc-nd/4.0/)

D status of children in Taiwan. Vitamin D is an essential element for the regulation of serum calcium and phosphate, and severe VDD is typically associated with overt hypocalcemia, hypophosphatemia, high alkaline phosphatase (ALP), and high intact parathyroid hormone (iPTH). Therefore, in this study, we aimed to investigate the total burden of VDD, and identify risk factors and determine the associations of calcium and ionized calcium, phosphorus, ALP and iPTH with VDD among children in Taiwan within a tertiary medical center setting.

2. METHODS

2.1. Setting, data sources, and study population

Information on the study cohort was extracted from the Big Data Center of Taipei Veterans General Hospital.¹² Comprehensive data from medical records were retrieved, including diagnostic tests, laboratory data, and medications of inpatient, outpatient, and emergency department visits. Diagnostic coding was based on the International Classification of Diseases, Tenth Revision, Clinical Modification (ICD-10-CM) codes. This retrospective study included all children (<18 years of age) who visited the outpatient or inpatient services at Taipei Veterans General Hospital from January 1, 2018, to December 31, 2023, and underwent 25-hydroxyvitamin D (25(OH)D) evaluations at either service for any clinical cause. This study was approved by the Taipei Veterans General Hospital Institutional Review Board and the need for informed consent was waived (2024-05-004BC).

2.2. Vitamin D level

Vitamin D status was assessed by measuring blood levels of total 25(OH)D using an electrochemiluminescence immunoassay. The date of the first 25(OH)D measurement was considered the index date. The level of 25(OH)D was classified as being sufficient (\geq 30 ng/mL), insufficient (\geq 20-30 ng/mL), or deficient (<20 ng/mL). Most experts agree that 25(OH)D levels below 20 ng/mL are generally too low for bone and overall health.¹³ Therefore, to analyze the risk factors for VDD, we divided the children into a non-VDD (sufficient/insufficient) group (\geq 20 ng/mL) and a VDD group (<20 ng/mL) based on the 25(OH)D level at first assessment. Repeat tests were analyzed separately from the first test.

2.3. Demographic and clinical data and comorbidity

Demographic characteristics, including age, sex, and residential location were extracted for analysis. Age was categorized into four groups: 0 to 1 years (infant), >1 to 6 years (preschool children), >6 to 12 years (school-age children), and >12 years (adolescents). To evaluate the impact of seasonal variability, the seasons during which the blood samples were collected were divided into two groups: spring (March, April, May)/ winter (December, January, February), and summer (June, July, August)/autumn (September, October, November). Residential locations were also categorized into two groups: areas with low sun exposure (<1400 hours annually), and areas with high sun exposure (≥1400 hours annually), based on the accumulated sun exposure throughout 2022 reported by the Central Weather (https://www.cwa.gov.tw/V8/C/C/Statistics/ Administration monthlydata.html).

Inflammatory markers including C-reactive protein (CRP) and white blood cell (WBC) count measured within 4 weeks before or after the 25(OH)D evaluations were collected. An elevated CRP level was defined as ≥ 0.5 mg/dL, while leukocytosis was defined as a WBC count >1.1 × 10⁴/mm³. Location where 25(OH)D were collected (outpatient or inpatient services), and the presence of acute respiratory tract infection based on corresponding ICD-10-CM codes (Supplementary Table 1, http://links.lww.com/JCMA/A322) were also ascertained.

www.ejcma.org

Author contributions: Dr. Yu-Chun Chen and Dr. Jei-Wen Chang contributed equally to the work.

Conflicts of interest: Dr. Yu-Chun Chen and Dr. Jei-Wen Chang, editorial board members at Journal of the Chinese Medical Association, had no role in the peer review process of or decision to publish this article. The other authors declare that they have no conflicts of interest related to the subject matter or materials discussed in this article.

doi: 10.1097/JCMA.000000000001235

Table 1

Demographic characteristics, clinical data, and comorbidities of the study population according to vitamin D status

	Total	Non-VDD	VDD	
Variable	(n = 1222)	(n = 625)	(n = 597)	р
Sex				< 0.001
Male	559 (45.7%)	328 (52,5%)	231 (38.7%)	
Female	663 (54.3%)	297 (47.5%)	366 (61.3%)	
Mean age (v)	8.5 ± 5.4	7.1 ± 5.1	9.9 ± 5.3	< 0.001
Age (v)				
0-1	168 (13.7%)	87 (13.9%)	81 (13.6%)	< 0.001
>1-6	274 (22.4%)	210 (33.6%)	64 (10.7%)	
>6-12	380 (31.1%)	195 (31.2%)	185 (31.0%)	
>12	400 (32.7%)	133 (21.3%)	267 (44.7%)	
Seasons of measurement				< 0.001
Summer/autumn	716 (58.6%)	405 (64.8%)	311 (52.1%)	
Sprina/winter	506 (41.4%)	220 (35.2%)	286 (47.9%)	
Residential location				0.674
Cities with high sun exposure (≥1400 h/v)	293 (24.0%)	153 (24.5%)	140 (23.5%)	
Cities with low sun exposure (<1400 h/y)	929 (76.0%)	472 (75.5%)	457 (76.5%)	
Acute inflammatory markers		× ,		
$CRP \ge 0.5 \text{ mg/dL}$	87/340 (25.6%)	36/340 (24.5%)	51/340 (26.4%)	0.685
CRP level (mg/dL)	0.7 ± 2.1	0.5 ± 1.0	0.8 ± 2.6	0.131
WBC count $>1.1 \times 10^4$ /mm ³	141/901 (15.6%)	83/901 (16.9%)	58/901 (14.2%)	0.269
WBC count (/mm ³)	8.1 ± 3.6	8.3 ± 3.6	7.8 ± 3.6	0.029
Location of measurement				< 0.001
Outpatient	910 (74.5%)	496 (79.4%)	414 (69.3%)	
Hospitalization	312 (25.5%)	129 (20.6%)	183 (30.7%)	
Acute respiratory tract infection	52 (4.3%)	34 (5.4%)	18 (3.0%)	0.036
Comorbidities	. ,			
Prematurity	130 (10.6%)	71 (11.4%)	59 (9.9%)	0.402
Congenital heart disease	68 (5.6%)	43 (6.9%)	25 (4.2%)	0.040
Malignancy	180 (14.7%)	98 (15.7%)	82 (13.7%)	0.338
Iron deficiency anemia	133 (10.9%)	80 (12.8%)	53 (8.9%)	0.028
Atopy	112 (9.2%)	60 (9.6%)	52 (8.7%)	0.590
Autoimmune disorders	19 (1.6%)	7 (1.1%)	12 (2.0%)	0.209
CKD and ESRD	4 (0.3%)	2 (0.3%)	2 (0.3%)	1.000ª
Nephrotic syndrome	6 (0.5%)	1 (0.2%)	5 (0.8%)	0.116ª
Liver cirrhosis	3 (0.2%)	2 (0.3%)	1 (0.2%)	1.000ª
Biliary atresia	6 (0.5%)	3 (0.5%)	3 (0.5%)	1.000ª
Autism	12 (1.0%)	9 (1.4%)	3 (0.5%)	0.097ª
Depression	8 (0.7%)	1 (0.2%)	7 (1.2%)	0.035ª

CKD = chronic kidney disease; CRP = C-reactive protein; ESRD = end-stage renal disease; VDD = vitamin D deficiency; WBC = white blood cell. *Fisher's exact test.

Comorbidities including prematurity, congenital heart disease, malignancy, atopy, autoimmune disorders, chronic kidney disease (CKD) or end-stage renal disease (ESRD), nephrotic syndrome, liver cirrhosis, biliary atresia, autism, and depression were extracted based on the presence of corresponding ICD-10-CM codes. The presence of iron deficiency anemia was determined according to relevant diagnostic codes and/or a transferrin saturation level (iron to total iron-binding capacity ratio) <20%. Details of the ICD-10-CM codes of comorbidities are shown in Supplementary Table 1 (http://links.lww.com/ JCMA/A322).

2.4. Biochemical parameters

Biochemical parameters measured within 3 months before the index date, including serum total calcium, ionized calcium, phosphorous, alkaline phosphatase (ALP), and intact parathyroid hormone (iPTH) were assessed. Hypocalcemia, hypophosphatemia, elevated ALP, and hyperparathyroidism were defined based on age- and sex-specific reference ranges of calcium, ionized calcium, phosphorous, ALP, and iPTH, respectively. Details

www.ejcma.org

of the reference normal ranges are shown in Supplementary Table 2 (http://links.lww.com/JCMA/A322).

2.5. Statistical analysis

Continuous data are presented as mean ± SD, and categorical data are presented as number (proportion). The χ^2 test or Fisher's exact test was used to compare categorical variables between the VDD and non-VDD groups. Potential clinical risk factors for VDD were initially evaluated using univariate logistic regression analysis. Factors with p values <0.05 in univariate analysis were then included in multivariate logistic regression, and odds ratios (ORs) and 95% CIs were calculated. To minimize analytical bias caused by small sample sizes and rare events, Firth's penalizedlikelihood logistic regression analyses were used to determine the biochemical predictors (calcium, phosphorus, ALP, iPTH) of VDD. The McNemar test was used to assess changes in vitamin D status between the index measurement and first repeat measurement of 25(OH)D. All statistical tests were two-sided, and a significance level of p < 0.05 was considered to indicate statistical significance. Statistical analyses were conducted utilizing

۲

Lin et al.



Fig. 1 Prevalence of VDD stratified by sex and age. In each age group, the prevalence of VDD in girls was higher compared with boys. The prevalence of VDD was highest in females aged >12 y (71.9%) and lowest in males aged >1-6 y (18.4%). VDD = vitamin D deficiency.

SPSS version 24.0 (SPSS Inc., Chicago, IL) and R software version 4.4.1 (R Core Team 2024).

3. RESULTS

3.1. Characteristics of the study population

The demographic characteristics of the study population based on serum 25(OH)D levels are summarized in Table 1. In total, 1222 children had serum 25(OH)D measurements during the study period, of whom 559 (45.7%) were male and 663 (54.3%) were female. The mean age of the participants was 8.5 ± 5.4 years. Regarding the age groups, 168 (13.7%), 274 (22.4%), 380 (31.1%), and 400 (32.7%) children were 0 to 1, >1 to 6, >6 to 12, and >12 years, respectively. The mean level of 25(OH)D was 21.4 \pm 9.9 ng/mL. Overall, the prevalence rates of vitamin D sufficiency, vitamin D insufficiency (VDI), and VDD were 12.4%, 38.8%, and 48.9%, respectively, with 25(OH)D concentrations of 39.7 \pm 11.9, 24.3 \pm 2.8 and 14.5 \pm 4.2 ng/mL, respectively.

3.2. Vitamin D status in different sex and age groups

Compared with the vitamin D sufficiency group, there were significantly higher percentages of females in the VDI (49.8% vs 40.4%) and VDD (61.3% vs 40.4%) groups. The distributions of vitamin D status according to age and sex are illustrated in Fig. 1. The prevalence of VDD was higher in the children aged >12 years (66.8%), followed by those aged >6 to 12 years (48.7%), 0 to 1 (48.2%), and >1 to 6 years (23.4%). The children with VDD were more likely to be older (9.9 ± 5.3 vs 7.1 ± 5.1 years, p < 0.001). There were significant associations between different vitamin D status and age groups (p < 0.001, Table 1). The highest prevalence of VDD (71.9%) was observed in females aged >12 years, and the lowest prevalence (18.4%) was observed in male children aged >1 to 6 years.

3.3. Residential location and season

Regarding the residential location, most of the children (76.0%) resided in areas with low sun exposure, and the others (24.0%) lived in areas with high sun exposure. There was no difference in vitamin D status between different residential locations according to total annual sun exposure. Regarding seasons, 506 (41.4%) children underwent serum vitamin D assessments in spring/winter, and 716 (58.6%) in summer/autumn. The mean

25(OH)D concentration in summer/autumn was 22.3 ± 10.2 ng/mL, compared with 20.2 ± 9.2 ng/mL in spring/winter, and the difference was statistically significant (p < 0.001). Among the children with VDD, more underwent 25(OH)D tests in spring/winter than in summer/autumn (47.9% vs 35.2%, p < 0.001).

3.4. Hospitalization, acute inflammatory markers, acute respiratory tract infection, and comorbidities

In our cohort, 74.5% of the 25(OH)D tests were ordered in outpatient clinics. Among the children who were hospitalized, only nine were admitted to the intensive care unit. The children in whom 25(OH)D was measured in outpatient clinics had a higher level than those who had measurements while hospitalized (22.1 \pm 9.5 vs 19.5 \pm 10.8 ng/mL, *p* < 0.001). The VDD group had a higher frequency of inpatient 25(OH)D measurements (30.7% vs 20.6%, p < 0.001). Three hundred and forty (27.8%) and 901 (73.7%) patients had CRP and WBC data, respectively. The average CRP level was $0.7 \pm 2.1 \text{ mg/dL}$, and there was no significant difference between the VDD group and non-VDD group $(0.8 \pm 2.6 \text{ vs } 0.5 \pm 1.0 \text{ mg/dL}, p = 0.131)$. The average WBC count was $8.1 \pm 3.6 \times 10^4$ /mm³, and the WBC count was significantly lower in the VDD group compared with the non-VDD group (7.8 \pm 3.6 vs 8.3 \pm 3.6 \times 10⁴/mm³, p = 0.029). However, neither elevated CRP level nor leukocytosis was associated with vitamin D status (p = 0.685 and 0.269, respectively). In addition, 3.0% of the VDD group and 5.4% of the non-VDD group had a respiratory tract infection 4 weeks before or after 25(OH)D measurement. The incidence of respiratory tract infection was significantly higher in the non-VDD group (p = 0.036)

The most common comorbidity was malignancy, followed by iron deficiency anemia and prematurity. The children with VDD had a much higher prevalence of depression (1.2% vs 0.2%, p= 0.035, Table 1). Conversely, iron deficiency anemia (8.9% vs 12.8%, p = 0.028) and congenital heart disease (4.2% vs 6.9%, p = 0.040) were less prevalent among the children with VDD.

3.5. Clinical predictors of VDD

The results of the univariate and multivariate logistic regression analyses are shown in Table 2. Univariate analysis showed that the female children, those aged >12 years, those who underwent 25(OH)D assessments during spring/winter, and those who were hospitalized were associated with VDD. The children aged >1 to 6 years, those diagnosed as acute respiratory infection, and

J Chin Med Assoc

Table 2

Logistic regression analysis of the clinical risk factors for vitamin D deficiency

	Univariate		Multivariate			
	OR	95% CI	р	OR	95% CI	р
Female	1.750	1.394-2.197	< 0.001	1.624	1.270-2.076	< 0.001
Age (y)						
0-1	1 (Reference)			1 (Reference)		
>1-6	0.327	0.217-0.494	< 0.001	0.391	0.244-0.628	< 0.001
>6-12	1.019	0.709-1.465	0.919	1.172	0.755-1.819	0.480
>12	2.156	1.494-3.113	< 0.001	2.479	1.584-3.881	< 0.001
Seasons of measurement						
Summer/autumn	1 (Reference)			1 (Reference)		
Spring/winter	1.693	1.346-2.130	< 0.001	1.716	1.340-2.197	< 0.001
Low sun exposure area (<1400 h/y)	0.945	0.717-1.229	0.674			
Acute inflammatory markers						
CRP ≥0.5 mg/dL	1.107	0.676-1.814	0.685			
WBC count >1.1 \times 10 ⁴ /mm ³	0.814	0.566-1.172	0.269			
Location of measurement						
Outpatient	1 (Reference)			1 (Reference)		
Hospitalization	1.700	1.310-2.205	< 0.001	1.949	1.439-2.640	< 0.001
Acute respiratory tract infection	0.540	0.302-0.968	0.038	0.849	0.456-1.580	0.605
Comorbidities						
Prematurity	0.856	0.594-1.233	0.403			
Congenital heart disease	0.592	0.357-0.981	0.042	0.574	0.317-1.040	0.067
Malignancy	0.856	0.623-1.176	0.338			
Iron deficiency anemia	0.664	0.460-0.958	0.029	0.711	0.614-1.395	0.925
Atopy	0.898	0.609-1.326	0.590			
Autoimmune disorders	1.811	0.708-4.631	0.215			
CKD and ESRD	1.047	0.147-7.457	0.963			
Nephrotic syndrome	5.270	0.614-45.243	0.130			
Liver cirrhosis	0.523	0.047-5.779	0.597			
Biliary atresia	1.047	0.211-5.209	0.955			
Autism	0.346	0.093-1.283	0.112			
Depression	7.403	0.908-60.355	0.061			

CKD = chronic kidney disease; CRP = C-reactive protein; ESRD = end-stage renal disease; OR = odds ratio; WBC = white blood cell.

Table 3

Comparisons of biochemical parameters between different vitamin D status groups

Variable	Total (%)	Non-VDD (%)	VDD (%)	p
Serum calcium/ionized calcium (n = 683)				0.087
Low	17 (2.5)	6 (1.6)	11 (3.6)	
Normal or high	666 (97.5)	374 (98.4)	292 (96.4)	
Serum phosphorus (n = 695)				0.040
Low	51 (7.3)	21 (5.5)	30 (9.6)	
Normal or high	644 (92.7)	361 (94.5)	283 (90.4)	
Serum ALP (n = 621)				0.004
High	61 (9.8)	22 (6.6)	39 (13.5)	
Normal or low	560 (90.2)	311 (94.3)	249 (86.5)	
iPTH (n = 140)				< 0.00
High	22 (15.7)	3 (4.1)	19 (28.8)	
Normal or low	118 (84.3)	71 (95.9)	47 (71.2)	

ALP = alkaline phosphatase; iPTH = intact parathyroid hormone; VDD = vitamin D deficiency.

those with comorbidities including congenial heart disease and iron deficiency anemia had a lower odds of VDD. However, there were no significant associations between elevated CRP level, leukocytosis, or other comorbidities including malignancy, atopy, prematurity, autoimmune disorders, CKD and ESRD, nephrotic syndrome, liver cirrhosis, biliary atresia, autism and depression with VDD. Multivariate logistic regression analyses revealed that female sex (OR = 1.624, 95% CI = 1.270-2.076, p

< 0.001), age >12 years (OR = 2.479, 95% CI = 1.548-3.881, p < 0.001), undergoing vitamin D assessments in spring/winter (OR = 1.716, 95% CI = 1.340-2.197, p < 0.001), and hospitalization (OR = 1.949, 95% CI = 1.439-2.640, p < 0.001) were associated with VDD. Compared with children aged 0 to 1 years, children aged >1 to 6 years were associated with decreased odds of VDD (OR = 0.391, 95% CI = 0.244-0.628, p < 0.001). The associations between acute respiratory tract infection, iron deficiency

www.ejcma.org

437

Lin et al.

Table 4 Firth's penalized-likelihood logistic regression analyses of calcium, phosphorus, ALP, and iPTH for vitamin D deficiency

	Univariate			Multivariate ^a		
	OR	95% CI	р	OR	95% CI	р
Hypocalcemia	2.265	0.855-6.000	0.100			
Hypophosphatemia	1.809	1.020-3.210	0.043	2.495	0.460-13.539	0.289
Elevated ALP	2.192	1.272-3.775	0.005	1.901	0.413-8.747	0.409
Hyperparathyroidism	8.386	2.537-27.718	< 0.001	8.667	1.338-56.157	0.024

ALP = alkaline phosphatase; iPTH = intact parathyroid hormone; OR = odds ratio.

^aAdjusted for age and sex.



Fig. 2 Transition in vitamin D status between the first and the consecutive tests in the study cohort. The Sankey diagram illustrates changes in vitamin D status among the 439 participants, categorized into sufficiency (vitamin D sufficiency), insufficiency (vitamin D insufficiency), and deficiency (vitamin D deficiency) groups based on the first and consecutive tests. The width of the flows corresponds to the number of participants transitioning between categories. Of the 439 participants, 41 were sufficient, 127 insufficient, and 271 deficient in the first test. In the consecutive test, 93 participants were sufficient, 167 insufficient, and 179 deficient.

anemia, and congenital heart disease with VDD disappeared after multivariate adjustments.

3.6. Biochemical predictors of VDD

The status of serum calcium/ionized calcium, phosphorous, ALP, and iPTH are shown in Table 3. Overall, 683 (55.9%), 695 (56.9%), 621 (50.8%), and 140 (11.5%) of the children had calcium/ionized calcium, phosphorous, ALP, and iPTH measurement data, respectively. Hypocalcemia, hypophosphatemia, elevated ALP, and elevated iPTH were noted in 11 (3.6%), 30 (9.6%), 39 (13.5%), and 19 (28.8%) of the children with VDD, respectively. Hypocalcemia, hypophosphatemia, elevated ALP, and hyperparathyroidism had specificities of 98.4%, 94.5%, 93.4%, and 95.9%, respectively, for the presence of VDD. There were significant differences in the proportion of hypophosphatemia, elevated ALP, and hyperparathyroidism between the VDD and non-VDD groups. Although a higher proportion of the VDD patients had hypocalcemia (3.6% vs 1.6%), the difference did not reach statistical significance. Univariate Firth's penalized-likelihood logistic regression analysis showed that hypophosphatemia, elevated ALP, and hyperparathyroidism were associated with a significantly increased odds of VDD (Table 4). However, multivariate Firth's penalized-likelihood

logistic regression analysis after adjusting for potentially confounding factors including age and sex showed that only elevated iPTH was independently associated with VDD (OR = 8.667, 95% CI = 1.338-56.157, p = 0.024).

3.7. Vitamin D supplementation and repeat assessments of serum 25(OH)D level

Overall, 394 (32.2%) of the children were given prescribed vitamin D supplements following the index test. In addition, 439 children underwent a 25(OH)D retest, with an average interval of 6.8 ± 7.3 months; 31.2% of the first retests were done within 3 months of the first test, and 32.6% were performed between 3 and 6 months after the first test. Forty-one (9.3%) of the retests were done in those who initially had vitamin D sufficiency. Among the children who underwent retests, 38.0% and 40.8% had VDI and VDD, respectively, and 222 (50.6%) were given prescribed vitamin D supplements. The changes in vitamin D status are summarized in Fig. 2. The 25(OH)D level in the retest was significantly higher than that in the initial test (24.1 \pm 11.9 vs 19.4 \pm 10.1 ng/mL, p < 0.001). At the second assessment, 93 (21.2%) children were categorized as having vitamin D sufficiency, 167 (38.0%) with VDI, and 179 (40.8%) with VDD. The vitamin D status improved in 150 patients (34.2%) on retest,

www.ejcma.org

however, there was no change or decline in vitamin D status in 241 (54.9%) and 48 (10.9%) children, respectively. Among the children who were retested, the proportion of those with VDD on retest significantly dropped (40.8% vs 61.7%, p < 0.001).

4. **DISCUSSION**

The concentration of serum total 25(OH)D is currently considered to be the best biomarker to assess vitamin D status. Based on a cut-off value of <20 ng/mL (50 nmol/L), VDD is a very pervasive condition worldwide especially people residing at high latitudes, particularly above 35°N and below 35°S.14 The prevalence of VDD (defined as 25(OH)D levels <50 nmol/L or 20 ng/mL) has been estimated to be 9.7% in children aged 1 to 11 years and 25% in children aged 12 to 19 years in the USA,¹⁵ and 26.7% to 35.7% in Canada.¹⁶ VDD is also common in Asia. In Japan, the prevalence rates of VDD and VDI among children aged 2 years have been estimated to be 25.4% and 50.9%, respectively.17 In Korea, 58.6% of children have been estimated to have VDD and 38.5% VDI.¹⁸ In addition, a multicenter hospital-based study in China reported a prevalence of VDD of 22.6%.¹⁹ Despite being an area with high sun exposure, the prevalence of VDD still varies from 0.9% to 96.4% in Southeast Asia.7 A meta-analysis study reported that the prevalence of VDD in children (<18 years of age) was 48.5%,²⁰ which is similar to the current study where 48.9% and 38.8% of the children had VDD and VDI, respectively. Furthermore, we found that VDD affected both male and female children irrespective of age, indicating the endemic nature of VDD in Taiwan, even though Taiwan is situated between 21°N and 25°N latitude.

Previous studies have shown that VDD is related to age. Consistent with most countries, we also found that adolescents were at risk of developing VDD. Notably, the prevalence of VDD observed among adolescents in this present hospital-based study, with rates reaching 65.7% among 12- to 15-year-olds and 69.9% among 15- to 18-year-olds, far surpasses the prevalence in a community setting, with 15.3% in 13- to 15-years-olds and 23.7% in 16- to 18-years-olds.²¹ These findings indicate that VDD is prevalent in Taiwan, especially among adolescents seeking health services.

In the present study, we found that the peak prevalence of VDD is in adolescent group (66.8%), compared with 48.7% in the school-age group, 48.2% in infant, and 23.4% in the preschool group. Furthermore, an older age was associated with an increased odds of VDD. Compared with infant group (age 0-1 year), the children aged >1 to 6 years had lowest risk for VDD (OR = 0.391), and adolescents aged >12 years had highest odds of having VDD (OR = 2.479). These results may be due to a lack of exposure to sunlight because of a greater time spent indoors on school work or excess screen time in adolescents.³

In the present study, 48.2% of the infants had VDD. This figure is higher than the 44.1% reported by Chen et al²² who investigated infants in northern Taiwan from 2012 to 2014, but lower than a systematic review conducted by Oktaria et al7 who reported a prevalence of VDD in Southeast Asian newborns ranging from 52% to 90%. In Taiwan, the rate of exclusive breastfeeding under 6 months of age was 44.8% in 2016 and 46.2% in 2018.23 As exclusively breastfed infants are at an increased risk of developing VDD, the American Academy of Pediatrics advises supplementing exclusively or partially breastfed infants with 400 IU/d of vitamin D from birth until they are able to consume over 1000 mL/d of vitamin D-fortified formula.²⁴ Preterm infants are vulnerable to VDD because of less vitamin D deposit and insufficient maternal vitamin D supply, intolerance to human milk formulas, malabsorption caused by necrotizing enterocolitis, and parenteral nutrition-associated cholestasis.²⁵ Supplementation of vitamin D ranges from 200 to

1000 IU/d for premature infants effectively reduced the risk of VDD. $^{26\-29}$

In this study, we also found that VDD was associated with female sex. A growing body of evidence has shown that VDD is more prevalent in obese individuals. Proposed causative factors include body fat acting as a reservoir for vitamin D storage thereby reducing vitamin D bioavailability, volumetric dilution, low dietary vitamin D intake, and insufficient physical activity.³⁰ Women tend to have more fat mass than men, which may explain the difference in the prevalence of VDD. In addition, differences in sunscreen use in males and females may also be another explanation.

Previous studies have shown the immunomodulatory effects of vitamin D on the innate and adaptive immune systems by modulating the expressions of antimicrobial peptides such as cathelicidin and inflammatory cytokines.³¹ In critically ill children, VDD is very common and strongly associated with worse clinical outcomes, including higher severity of illness scores, use of vasopressors, fluid resuscitation, and mechanical ventilation support.^{32,33} Another study conducted in Spain showed that patients in pediatric intensive care units had twice the rate of VDD compared with healthy controls (29.5% vs 15.6%).³⁴ Hemodilution, interstitial extravasation, and decreased serum vitamin D-binding protein can cause rapid reductions and changes in serum 25(OH)D level during critical illness.35 Furthermore, clinical studies have suggested an association between VDD and an increased risk of respiratory tract infection.^{31,36} In our study, we observed no significant association between VDD and inflammatory markers (CRP and leukocytosis); however, the children in whom 25(OH)D was measured during hospitalization had a 1.949-fold increased odds of VDD compared with those assessed in outpatient clinics. In contrast to other studies, we did not find an association between VDD and acute respiratory tract infection. We did not explore the prevalence of VDD in the critically ill children, as only a very small number of patients required intensive care in this study. Vitamin D plays an important pleiotropic role on different organs in the setting of critical illness, and further prospective studies are warranted to explore this issue.

VDD has been linked to numerous comorbid diseases and health conditions. Even though previous studies have reported associations between VDD with atopy,³⁷ autoimmune disorders,⁵ chronic liver diseases,³⁸ CKD or ESRD,⁵ nephrotic syndrome,⁵ congenital heart disease,³⁹ anemia,⁴⁰ depression,⁵ and autism,² these associations were not found in the present study.

Solar ultraviolet B is the major source of vitamin D in addition to dietary intake. An Australian study reported that sunlight exposure to the face, arms, or legs for approximately 10 to 15 minutes between 10:00 and 15:00 without the use of sunscreen at least three times a week was sufficient to induce vitamin D synthesis.⁴¹ Based on the important contribution of sunlight exposure to vitamin D levels, serum 25(OH)D may be a proxy for the amount of sun exposure. There is currently no consensus on the impact of seasonal changes on vitamin D status. The results of this study are in agreement with several published studies,3 in that spring/winter was associated with a greater odds of VDD and lower mean vitamin D levels compared with summer/autumn. However, some studies have reported no seasonal variation in VDD.⁴² Kimlin et al¹⁴ reported that during winter months, more ultraviolet is required to maintain vitamin D sufficiency only for those living at high latitudes, but not for those at latitudes below 25°N, such as Taiwan. Taiwan experiences four distinct seasons. In winter, the average temperature ranges from 14°C to 20°C with little rainfall. It is thus possible that people wear more clothing and participate less in outdoor activities, leading to decreased cutaneous vitamin D synthesis, and consequently the observed seasonal variations.

()

۲

Lin et al.

In our study, the high prevalence of VDD may also be attributable to low consumption of vitamin D-rich foods in addition to excess screen time and changes in lifestyle. Since dietary sources are unlikely to be sufficient, especially for infants and vegetarians, vitamin D supplements are often necessary to prevent VDD. The recommended dosage of vitamin D supplements to treat VDD depends on the severity of VDD and age of the patient.43 In obese patients, higher doses of vitamin D are necessary to treat VDD, and it is reasonable to set therapeutic goals of treatment to achieve a 25(OH)D level of >50 to 75 nmol/L or 20 to 30 ng/mL to relieve symptoms and decrease the risk of fractures. During the first month of supplementation, 25(OH)D levels have been reported to rise rapidly, and then to continue to rise slowly over the following 2 to 3 months.⁴⁴ Given the low risk of vitamin D toxicity such as hypercalcemia associated with excess vitamin D supplementation,⁴⁵ a repeat blood test for 25(OH)D is only suggested when clinically indicated and after 3 to 6 months of treatment. In the present study, 59.2% of the patients who had a repeat 25(OH)D test reached the threshold of >20 ng/mL. However, we did not have information on overthe-counter use of vitamin D supplements, dosage, adherence, or lifestyle modifications after the index 25(OH)D test. This restricts our ability to ascertain the impact of vitamin D supplements on changes in vitamin D status over time.

Low 25(OH)D status reduces the efficiency of intestinal calcium absorption, leading to an increase in iPTH level to restore calcium homeostasis. Secondary hyperparathyroidism mobilizes calcium from the skeleton and increases phosphorus wasting in the kidneys. The correlation between vitamin D level and ALP and iPTH and even the potential use as a screening test for VDD has been reported,^{22,46} however other studies have not shown any correlations between these biochemical parameters and VDD.⁴⁷ In accordance with a study conducted by Hashemipour et al,47 most of the children with VDD with available biochemical profile data in our study had normal calcium, phosphorus, ALP, and iPTH, indicating that the use of biochemical parameters alone has low sensitivity to detect mild VDD. Therefore, normal biochemical parameters may not be suitable as a screening test to rule out VDD. The multivariate analysis in this study showed that elevated iPTH was associated with an 8.667-fold increased odds of VDD.

To the best of our knowledge, this is the first large-scale hospital-based study of vitamin D status in children aged 0 to 18 years living in Taiwan, a low-latitude region (25°N) in East Asia. Our findings add to the knowledge of the epidemiology of VDD in countries situated at subtropical latitudes. However, there are several limitations. First, our study cohort was comprised of children seeking medical care at a tertiary care hospital for various reasons or conditions, and thus cannot be regarded as representative of the general population. It is also important to note that low levels of 25(OH)D are common during acute illnesses,31-33 which may lead to the overestimation of the prevalence of VDD in healthy children. Second, we lacked information on indoor/outdoor activity levels and clothing, which are crucial factors affecting cutaneous vitamin D synthesis. Third, we lacked data on the dietary intake of vitamin D, calcium, phosphorus, and over-the-counter vitamin D supplements. Fourth, some data were missing due to the retrospective design of our research, including on acute inflammatory markers and biochemical profile. In particular, missing data on iPTH could have led to bias and affected the results of the study. In addition, of the children with 25(OH)D measurements, only 904 (74.0%) and 761 (62.3%) had at least one acute inflammatory marker or biochemical profile measurement, respectively. Fifth, details of symptoms and body mass index, both of which may be important potential confounding factors in this study, were not available. In addition, the assessment of comorbidities relied

on the documentation of relevant ICD-10-CM codes in our hospital records, which may have underestimated the prevalence of comorbidities. We suggest that future prospective studies are needed to better understand the risk factors associated with VDD and improve the robustness of our results.

In conclusion, our study underscores that VDD was prevalent among Taiwanese children seeking various medical treatments in a hospital setting despite high sun exposure in Taiwan. Our findings highlight that hospitalized children, female, children aged greater than 12 years, testing during the spring/winter are associated with a higher likelihood of VDD. Physicians should be aware of VDD, especially in children at higher risk.

ACKNOWLEDGMENTS

We thank the Big Data Center of Taipei Veterans General Hospital (BDC, TPEVGH) and grant support from Taipei Veterans General Hospital (V113C-135).

APPENDIX A. SUPPLEMENTARY DATA

Supplementary data related to this article can be found at http://links.lww.com/JCMA/A322.

REFERENCES

- 1. Bikle DD. Extraskeletal actions of vitamin D. Ann N Y Acad Sci 2016;1376:29-52.
- Antonucci R, Locci C, Clemente MG, Chicconi E, Antonucci L. Vitamin D deficiency in childhood: old lessons and current challenges. J Pediatr Endocrinol Metab 2018;31:247–60.
- Hu Y, Jiang S, Lu J, Yang Z, Yang X, Yang L. Vitamin D status for Chinese children and adolescents in CNNHS 2016–2017. Nutrients 2022;14:4928.
- Robien K, Oppeneer SJ, Kelly JA, Hamilton-Reeves JM. Drug–vitamin D interactions: a systematic review of the literature. *Nutr Clin Pract* 2013;28:194–208.
- 5. Holick MF. Vitamin D deficiency. N Engl J Med 2007;357:266-81.
- Van Schoor N, Lips P. Global overview of vitamin D status. Endocrinol Metab Clin North Am 2017;46:845–70.
- Oktaria V, Putri DAD, Ihyauddin Z, Julia M, Sulistyoningrum DC, Koon PB, et al. Vitamin D deficiency in South-East Asian children: a systematic review. *Arch Dis Child* 2022;107:980–7.
- Won JW, Jung SK, Jung IA, Lee Y. Seasonal changes in vitamin D levels of healthy children in mid-latitude, Asian urban area. *Pediatr Gastroenterol Hepatol Nutr* 2021;24:207–17.
- Octavius GS, Shakila A, Meliani M, Halim A. Vitamin D deficiency is a public health emergency among Indonesian children and adolescents: a systematic review and meta-analysis of prevalence. *Ann Pediatr Endocrinol Metab* 2023;28:10–9.
- Puri S, Marwaha RK, Agarwal N, Tandon N, Agarwal R, Grewal K, et al. Vitamin D status of apparently healthy schoolgirls from two different socioeconomic strata in Delhi: relation to nutrition and lifestyle. *Br J Nutr* 2008;99:876–82.
- Mori H. Secular changes in relative height of children in Japan, South Korea and Taiwan: is "genetics" the key determinant? *Am J Biomed Sci* & *Res* 2019;4:AJBSR.MS.ID.000857.
- 12. Kuan AS, Chen TJ. Healthcare data research: the inception of the Taipei Veterans General Hospital big data center. J Chin Med Assoc 2019;82:679.
- Gallagher JC, Sai AJ. Vitamin D insufficiency, deficiency, and bone health. J Clin Endocrinol Metab 2010;95:2630–3.
- 14. Kimlin MG, Olds WJ, Moore MR. Location and vitamin D synthesis: is the hypothesis validated by geophysical data? *J Photochem Photobiol B* 2007;86:234–9.
- 15. Schleicher RL, Sternberg MR, Looker AC, Yetley EA, Lacher DA, Sempos CT, et al. National estimates of serum total 25-hydroxyvitamin D and metabolite concentrations measured by liquid chromatographytandem mass spectrometry in the US population during 2007–2010. J Nutr 2016;146:1051–61.

www.ejcma.org

Original Article. (2025) 88:6

- Sarafin K, Durazo-Arvizu R, Tian L, Phinney KW, Tai S, Camara JE, et al. Standardizing 25-hydroxyvitamin D values from the Canadian health measures survey. *Am J Clin Nutr* 2015;102:1044–50.
- Yang L, Sato M, Saito-Abe M, Irahara M, Nishizato M, Sasaki H, et al. 25-Hydroxyvitamin D levels among 2-year-old children: findings from the Japan Environment and Children's Study (JECS). *BMC Pediatr* 2021;21:1–8.
- Chung IH, Kim HJ, Chung S, Yoo EG. Vitamin D deficiency in Korean children: prevalence, risk factors, and the relationship with parathyroid hormone levels. *Ann Pediatr Endocrinol Metab* 2014;19:86–90.
- Yang C, Mao M, Ping L, Yu D. Prevalence of vitamin D deficiency and insufficiency among 460,537 children in 825 hospitals from 18 provinces in mainland China. *Medicine (Baltimore)* 2020;99:e22463.
- Cui A, Zhang T, Xiao P, Fan Z, Wang H, Zhuang Y. Global and regional prevalence of vitamin D deficiency in population-based studies from 2000 to 2022: a pooled analysis of 7.9 million participants. *Front Nutr* 2023;10:1070808.
- Pan WH. The Nutrition and Health Survey in Taiwan (NAHSIT) 2017-2020 report. *Ministry of Health and Welfare* 2022. Available at https:// www.hpa.gov.tw/EngPages/Detail.aspx?nodeid=3999&pid=15562. Accessed April 5, 2024.
- 22. Chen CM, Mu SC, Chen YL, Tsai LY, Kuo YT, Cheong IM, et al. Infants' vitamin D nutritional status in the first year of life in northern Taiwan. *Nutrients* 2020;12:404.
- 23. Ministry of Health and Welfare, R.O.C (Taiwan). Current status of breastfeeding in Taiwan. Ministry of Health and Welfare 2019. Available at https://www.hpa.gov.tw/Pages/Detail.aspx?nodeid=506&pid=463. Accessed April 5, 2024.
- 24. Wagner CL, Greer FR; American Academy of Pediatrics Section on Breastfeeding. American Academy of Pediatrics Section on Breastfeeding; American Academy of Pediatrics Committee on Nutrition. Prevention of rickets and vitamin D deficiency in infants, children, and adolescents. *Pediatrics* 2008;122:1142–52.
- Abrams SA. Vitamin D in preterm and full-term infants. Ann Nutr Metab 2020;76:6–14.
- Zittermann A, Pilz S, Berthold HK. Serum 25-hydroxyvitamin D response to vitamin D supplementation in infants: a systematic review and metaanalysis of clinical intervention trials. *Eur J Nutr* 2020;59:359–69.
- 27. Nehra D, Carlson SJ, Fallon EM, Kalish B, Potemkin AK, Gura KM, et al; American Society for Parenteral and Enteral Nutrition. ASPEN clinical guidelines: nutrition support of neonatal patients at risk for metabolic bone disease. *JPEN J Parenter Enteral Nutr* 2013;37:570–98.
- Abrams SA; Committee on Nutrition. Committee on Nutrition. Calcium and vitamin D requirements of enterally fed preterm infants. *Pediatrics* 2013;131:e1676–83.
- Agostoni C, Buonocore G, Carnielli VP, Curtis MD, Darmaun D, Decsi T, et al. Enteral nutrient supply for preterm infants: commentary from the European Society of Paediatric Gastroenterology, Hepatology and Nutrition Committee on Nutrition. J Pediatr Gastroenterol Nutr 2010;50:85–91.
- Muscogiuri G, Barrea L, Somma CD, Laudisio D, Salzano C, Pugliese G, et al. Sex differences of vitamin D status across BMI classes: an observational prospective cohort study. *Nutrients* 2019;11:3034.

- J Chin Med Assoc
- Bishop EL, Ismailova A, Dimeloe S, Hewison M, White JH. Vitamin D and immune regulation: antibacterial, antiviral, anti-inflammatory. *JBMR Plus* 2020;5:e10405.
- 32. Madden K, Feldman HA, Smith EM, Gordon CM, Keisling SM, Sullivan RM, et al. Vitamin D deficiency in critically ill children. *Pediatrics* 2012;130:421–8.
- McNally JD, Menon K, Chakraborty P, Fisher L, Williams KA, Al-Dirbashi OY, et al; Canadian Critical Care Trials Group. The association of vitamin D status with pediatric critical illness. *Pediatrics* 2012;130:429–36.
- 34. Rey C, Sánchez-Arango D, López-Herce J, Martínez-Camblor P, García-Hernández I, Prieto B, et al. Vitamin D deficiency at pediatric intensive care admission. *J Pediatr (Rio J)* 2014;90:135–42.
- Quraishi SA, Camargo CA, Jr. Vitamin D in acute stress and critical illness. Curr Opin Clin Nutr Metab Care 2012;15:625–34.
- Esposito S, Lelii M. Vitamin D and respiratory tract infections in childhood. BMC Infect Dis 2015;15:487.
- Kim MJ, Kim S-N, Lee YW, Choe YB, Ahn KJ. Vitamin D status and efficacy of vitamin D supplementation in atopic dermatitis: a systematic review and meta-analysis. *Nutrients* 2016;8:789.
- Pop TL, Sîrbe C, Benţa G, Mititelu A, Grama A. The role of vitamin D and vitamin D binding protein in chronic liver diseases. *Int J Mol Sci* 2022;23:10705.
- Abou Zahr R, Faustino EVS, Carpenter T, Kirshbom P, Hall EK, Fahey JT, et al. Vitamin D status after cardiopulmonary bypass in children with congenital heart disease. J Intensive Care Med 2017;32:508–13.
- Uwaezuoke SN. Vitamin D deficiency and anemia risk in children: a review of emerging evidence. *Pediatric Health Med Ther* 2017;8:47-55.
- 41. Samanek AJ, Croager EJ, Gies P, Milne E, Prince R, McMichael AJ, et al; Skin cancer prevention. Estimates of beneficial and harmful sun exposure times during the year for major Australian population centres. *Med J Aust* 2006;184:338–41.
- 42. Heidari B, Mirghassemi MBH. Seasonal variations in serum vitamin D according to age and sex. *Caspian J Intern Med* 2012;3:535.
- Dominguez LJ, Farruggia M, Veronese N, Barbagallo M. Vitamin D sources, metabolism, and deficiency: available compounds and guidelines for its treatment. *Metabolites* 2021;11:255.
- Vieth R. Vitamin D supplementation, 25-hydroxyvitamin D concentrations, and safety. Am J Clin Nutr 1999;69:842–56.
- 45. Genzen JR, Gosselin JT, Wilson TC, Racila E, Krasowski MD. Analysis of vitamin D status at two academic medical centers and a national reference laboratory: result patterns vary by age, gender, season, and patient location. *BMC Endocr Disord* 2013;13:1–14.
- 46. Ranathunga I, Naveenkumar JP, Sumanatilleke M, Somasundaram N. Prediction of 25-hydroxy vitamin D status by serum alkaline phosphatase (ALP) value. In: Proceedings from the European Congress of Endocrinology 2022, May 21–24, 2022; Milan, Italy, Eposter 183.
- Hashemipour S, Larijani B, Adibi H, Sedaghat M, Pajouhi M, Bastan-Hagh MH, et al. The status of biochemical parameters in varying degrees of vitamin D deficiency. J Bone Miner Metab 2006;24:213–8.

www.ejcma.org

()